Author's response to reviews

Title: What matters most to people with multiple myeloma? A qualitative study of views on quality of life.

Authors:

Thomas R Osborne (thomas.osborne@kcl.ac.uk)
Christina Ramsenthaler (christina.ramsenthaler@kcl.ac.uk)
Susanne de Wolf-Linder (susanne.1.de_wolf@kcl.ac.uk)
Stephen A Schey (sschey@nhs.net)
Richard J Siegert (richard.siegert@aut.ac.nz)
Polly M Edmonds (polly.edmonds@nhs.net)
Irene J Higginson (irene.higginson@kcl.ac.uk)

Version: 2 Date: 20 March 2014

Author's response to reviews:

Dear Editor,

Please find enclosed a revised manuscript with the following changes:

1) Line numbering added as requested

2) Figure legends and formatting. We have removed Figure 1 from the main body of the text leaving only the figure title at the end of the document. We have reduced the Figure 1 title to within the 15 word limit. There is no detailed legend for Figure 1.

Please accept our apologies for these omissions.

Yours Sincerely,

Dr Thomas Osborne